Author's response to reviews

Title: Behaviour Sequelae Following Acute Kawasaki Disease

Authors:

Daniel Carlton-Conway (Robert.Tulloh@ubht.swest.nhs.uk)
Raju Ahluwalia (Robert.Tulloh@ubht.swest.nhs.uk)
Lucy Henry (Lucy.Henry@kcl.ac.uk)
Colin Michie (colinm@easynet.co.uk)
Louise Wood (Robert.Tulloh@ubht.swest.nhs.uk)
Robert M Tulloh (Robert.Tulloh@ubht.swest.nhs.uk)

Version: 3 Date: 25 April 2005

Author's response to reviews:

Dear Dr Moylan

Thank you for inviting us to complete the formatting of our paper. The items are addressed in turn:
1. Abstract This is now formatted as requested in the word documents with the appropriate headings, minimal use of abbreviations and word count less than 350 words
2. Headings are correctly formatted and unnecessary capitals are removed
3. The reference format has been corrected
4. The tables are inserted with headings after the text
5. Kawasaki Disease has been rewritten Kawasaki disease throughout

We hope the manuscript is now acceptable for publication

With regards

Dr Robert Tulloh